Author's response to reviews

Title: Identification of an autoantibody panel to separate lung cancer from smokers and nonsmokers

Authors:

William N Rom (william.rom@nyumc.org)
Judith D Goldberg (jd.goldberg@nyumc.org)
Doreen Addrizzo-Harris (doreen.addrizzo@nyumc.org)
Heather N Watson (heather.watson@nyumc.org)
Michael Khilkin (michael.khilkin@nyumc.org)
Alissa K Greenberg (alissa.greenberg@nyumc.org)
David P Naidich (david.naidich@med.nyu.edu)
Bernard Crawford (bernard.crawford@nyumc.org)
Ellen Eylers (ellen.eylers@nyumc.org)
Daorong Liu (dliu@scripps.edu)
Eng M Tan (emtan@scripps.edu)

Version: 10 Date: 12 May 2010

Author's response to reviews:

Revisions have been made per the email sent on May 12, 2010 to Dr. William Rom.